Author's response to reviews

Title: Treatment with gefitinib after erlotinib-induced liver injury: a case report

Authors:

Katsumi Nakatomi (nakatomi@nagasaki-u.ac.jp)
Yoichi Nakamura (yi-nakamu@umin.ac.jp)
Tetsuya Iida (tetsu1@mocha.ocn.ne.jp)
Shigeru Kohno (s-kohno@nagasaki-u.ac.jp)

Version: 2 Date: 1 September 2011

Author's response to reviews:

We modified our manuscript according to your advice as below. Please receive our revised version.

1. As the patient is still alive after the treatment, please explain in the consent section why you have obtained the written consent from the family of the patient and not the patient herself.
   Ans. This patient has already died of lung cancer, so we add this statement in page 5 line 3.

2. Please include patient’s ethnicity in the abstract and case presentation section.
   Ans. We added ethnicity (Japanese) in the abstract and case presentation section.

Thank you in advance for considering this case study for publication in The Journal of Medical Case Reports and I look forward to hearing from you in due course.

Yours sincerely,

Dr Yoichi Nakamura